Author's response to reviews

Title: Public Perceptions and Attitudes toward Thalassaemia: Influencing factors in a Multi-racial Population

Authors:

- Li Ping Wong (wonglp@ummc.edu.my)
- Elizabeth George (elizg@medic.upm.edu.my)
- Jin-Ai Mary Anne Tan (maryanne@ummc.edu.my)

Version: 6 Date: 18 March 2011

Author's response to reviews: see over
Dear Editor BMC Public Health,

Enclosed is the revision in blue highlights.

Public Perceptions and Attitudes toward Thalassaemia: Influencing factors in a Multi-racial Population

Li Ping, Wong\textsuperscript{a,b,c-§}, Elizabeth, George\textsuperscript{d}, Jin-Ai Mary Anne, Tan\textsuperscript{e-§}
\textsuperscript{a} Department of Social and Preventive Medicine Faculty of Medicine, University of Malaya, Kuala Lumpur, Malaysia
\textsuperscript{b} Julius Centre, University of Malaya, Kuala Lumpur, Malaysia
\textsuperscript{c} Centre for Population Health, University of Malaya, Kuala Lumpur, Malaysia
\textsuperscript{d} Hematology Unit, Department of Pathology, Faculty of Medicine and Health Sciences, Universiti Putra Malaysia (UPM), Serdang, Selangor, Malaysia
\textsuperscript{e} Department of Molecular Medicine, Faculty of Medicine, University of Malaya, 50603, Kuala Lumpur, Malaysia